CSE 321: Discrete Structures Winter 2000

Course Information

Lectures: MWF 11:30-12:20 LOW 101 Instructor: Josh Benaloh benaloh@microsoft.com (425) 703-3871 Office Hour: Mon 12:30-2:00 Sieg 226C

Sections: Thurs 1:30-2:30 EEB 222 Thurs 2:30-3:20 LOW 222 TA: Erik Vee Sieg 429 env@cs.washington.edu 543-5129

Prerequisites: Math 126 and CSE 143.

Textbook: Kenneth H. Rosen, Discrete Mathematics and its Applications, Fourth Edition, McGraw-Hill, 1999.

Grading: The course grade will be based on homework (40%), a midterm exam (20%), and a final exam (40%). All percentages are approximate. The final exam is scheduled for Wednesday, March 15, 2000 at 2:30 - 4:20 p.m.

Homework: In general, homework assignments will be distributed on Fridays and due at the beginning of class on the following Wednesdays. Solutions will be discussed in class the day assignments are due and in greater detail in sections the following day. A consequence of this schedule is that no late homework will be accepted!

Course Web: All handouts and a log of all mail sent to the class mailing list will also be available on the course web at the following URL:

http://www.cs.washington.edu/education/courses/321/CurrentQtr/

Mailing List: The class mailing list is cse321@cs.washington.edu. Instructions for subscribing to the class mailing list can be found on the course web.

We will use this list for announcements of general interest to the class. Students should also feel free to use it to ask questions, post information, or initiate discussions of general interest to the class. Questions or comments that are not of general interest should instead be directed to the TA or instructor at the addresses above.

If you would like to request academic accommodations due to a disability, please contact Disabled Student Services, 448 Schmitz, 543–8924 (V/TDD). If you already have a letter from Disabled Student Services indicating you have a disability that requires academic accommodations, please let the instructor know so we can discuss the appropriate accommodations.